Author's response to reviews


Authors:

Paul A Batchelor (paulb@public-health.ucl.ac.uk)
Aubrey Sheiham (a.sheiham@ucl.ac.uk)

Version: 3 Date: 18 November 2005

Author's response to reviews:

The previous modified submission contained an error in the text. The description of the group according to water fluoridation and sealant status is incorrect in the text; the figures are labeled correctly. The text in the main paper has been modified to address this shortfall. Please accept our apologies.